Author's response to reviews

Title: Record linkage to obtain birth outcomes for the evaluation of screening biomarkers in pregnancy: a feasibility study

Authors:

Samantha J Lain (samlain@med.usyd.edu.au)
Charles S Algert (calgert@med.usyd.edu.au)
Vitomir Tasevski (vtasevsk@nsccahs.health.nsw.gov.au)
Jonathan M Morris (jmorris@med.usyd.edu.au)
Christine L Roberts (clroberts@med.usyd.edu.au)

Version: 3 Date: 8 July 2009

Author's response to reviews:

Table 3 was in Landscape format.

We have divided this table into two A4 portrait tables, and numbered them as Tables 3A and 3B. The text has been changed so that it now refers to these as Tables 3A and 3B. We have also removed the vertical lines from the tables. The data contained in the rows and columns are unchanged, except for formatting of line space and font size.

All changes have been accepted, and underlining removed.